Author's response to reviews

Title: Gaseous air pollution and emergency hospital visits for hypertension in Beijing, China: a time-stratified case-crossover study

Authors:

Yuming Guo (ymguo@yahoo.cn)
Shilu Tong (s.tong@qut.edu.au)
Shanshan Li (shandy0601@yahoo.com.cn)
Adrian G Barnett (a.barnett@qut.edu.au)
Weiwei Yu (weiwei.yu@qut.edu.au)
Yanshen Zhang (zhangyanshen@126.com)
Xiaochuan Pan (xcpan@hsc.pku.edu.cn)

Version: 6 Date: 29 September 2010

Author's response to reviews:

We made some small typographical changes in the revision.